Author's response to reviews

Title: Diagnostic utility of a one-item question to screen for depressive disorders: results from the MONICA/KORA study

Authors:

Eva Blozik (e.blozik@uke.de)  
Martin Scherer (m.scherer@uke.de)  
Maria E Lacruz (elena_kt@yahoo.co.uk)  
Karl-Heinz Ladwig (ladwig@helmholtz-muenchen.de)

Version: 2 Date: 1 July 2013

Author's response to reviews:

Dear Members of the Editorial Board,

Please find attached our revised original paper entitled “Diagnostic utility of a one-item question to screen for depressive disorders: results from the MONICA/KORA study” (MS: 1273535805102159).

As requested, you find a statement that written informed consent was obtained from each study participant and that the study was approved by the ethics committee of the Bavarian Medical Association on page 5 (marked in light grey).

We thank you in advance for your further processing of the manuscript. Please come back to us should there be any additional problems or questions.

Yours sincerely,
Eva Blozik